

# 

**Citation:** Koolhof IS, Beeton N, Bettiol S, Charleston M, Firestone SM, Gibney K, et al. (2024) Testing the intrinsic mechanisms driving the dynamics of Ross River Virus across Australia. PLoS Pathog 20(2): e1011944. https://doi.org/ 10.1371/journal.ppat.1011944

Editor: Alexander E. Gorbalenya, Leiden University Medical Center: Leids Universitair Medisch Centrum, NETHERLANDS

Received: June 1, 2023

Accepted: January 4, 2024

Published: February 15, 2024

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.ppat.1011944

**Copyright:** © 2024 Koolhof et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** The data underlying the results presented in the study are available from the following sources: RRV Human

RESEARCH ARTICLE

# Testing the intrinsic mechanisms driving the dynamics of Ross River Virus across Australia

lain S. Koolhof<sup>1,2\*</sup>, Nicholas Beeton<sup>3</sup>, Silvana Bettiol<sup>1</sup>, Michael Charleston<sup>2</sup>, Simon M. Firestone<sup>4</sup>, Katherine Gibney<sup>5</sup>, Peter Neville<sup>6</sup>, Andrew Jardine<sup>6</sup>, Peter Markey<sup>7</sup>, Nina Kurucz<sup>7</sup>, Allan Warchot<sup>7</sup>, Vicki Krause<sup>7</sup>, Michael Onn<sup>8</sup>, Stacey Rowe<sup>5</sup>, Lucinda Franklin<sup>5</sup>, Stephen Fricker<sup>9</sup>, Craig Williams<sup>9</sup>, Scott Carver<sup>2,10,11</sup>

1 College of Health and Medicine, Tasmanian School of Medicine, University of Tasmania, Hobart, Tasmania, Australia, 2 College of Sciences and Engineering, School of Natural Sciences, University of Tasmania, Hobart, Tasmania, Australia, 3 Data61, CSIRO, Hobart, Tasmania, Australia, 4 Melbourne Veterinary School, Faculty of Science, University of Melbourne, Melbourne, Victoria, Australia, 5 Victorian Department of Health and Human Services, Communicable Disease Epidemiology and Surveillance, Health Protection Branch, Melbourne, Victoria, Australia, 6 Department of Health, Western Australia, Environmental Health Directorate, Public and Aboriginal Health Division, Perth, Western Australia, Australia, 7 Centre for Disease Control, Northern Territory Department of Health, Northern Territory, Darwin, Australia, 8 Brisbane City Council, Brisbane, Queensland, Australia, 9 Australian Centre for Precision Health, University of South Australia, Adelaide, South Australia, Australia, 10 Odum School of Ecology, University of Georgia, Georgia, United States of America, 11 Center for the Ecology of Infectious Diseases, University of Georgia, Georgia, United States of America

\* koolhofi@utas.edu.au

# Abstract

The mechanisms driving dynamics of many epidemiologically important mosquito-borne pathogens are complex, involving combinations of vector and host factors (e.g., species composition and life-history traits), and factors associated with transmission and reporting. Understanding which intrinsic mechanisms contribute most to observed disease dynamics is important, yet often poorly understood. Ross River virus (RRV) is Australia's most important mosquito-borne disease, with variable transmission dynamics across geographic regions. We used deterministic ordinary differential equation models to test mechanisms driving RRV dynamics across major epidemic centers in Brisbane, Darwin, Mandurah, Mildura, Gippsland, Renmark, Murray Bridge, and Coorong. We considered models with up to two vector species (Aedes vigilax, Culex annulirostris, Aedes camptorhynchus, Culex globocoxitus), two reservoir hosts (macropods, possums), seasonal transmission effects, and transmission parameters. We fit models against long-term RRV surveillance data (1991-2017) and used Akaike Information Criterion to select important mechanisms. The combination of two vector species, two reservoir hosts, and seasonal transmission effects explained RRV dynamics best across sites. Estimated vector-human transmission rate (average  $\beta$  = 8.04x10<sup>-4</sup>per vector per day) was similar despite different dynamics. Models estimate 43% underreporting of RRV infections. Findings enhance understanding of RRV transmission mechanisms, provide disease parameter estimates which can be used to guide future research into public health improvements and offer a basis to evaluate mitigation practices.

Notification Data: The Department of Health and the Western Australian Notifiable Infectious Diseases Database (https://www.health.wa.gov. au/~/media/Files/Corporate/general% 20documents/Performance/Information% 20access/Information-Release-Form.xls); the Northern Territory Department of Health and Centre of Disease Control Surveillance Unit (https:// health.nt.gov.au/data-and-research/health-data/ health-data?SQ\_PAINT\_LAYOUT=664820&SQ\_ ASSET CONTENTS=&curr=578946&print=ves); the Queensland Department of Health and the Communicable Disease Branch (https://www. health.qld.gov.au/clinical-practice/guidelinesprocedures/diseases-infection/surveillance/reports/ notifiable/data-request); the Victorian Department of Health and the Communicable Disease Epidemiology and Surveillance Branch (https:// www.health.vic.gov.au/infectious-diseases/aboutinfectious-diseases-surveillance-in-victoria); the Department of Health South Australia and the Communicable Disease Control Branch (https:// www.sahealth.sa.gov.au/wps/wcm/connect/public +content/sa+health+internet/about+us/department +for+health+and+wellbeing/health+regulation+and +protection/communicable+disease+control +branch/communicable+disease+control+branch). Mosquito Surveillance Data: The Arbovirus Surveillance and Research Laboratory for Western Australia (https://www.health.wa.gov.au/articles/a e/arbovirus-surveillance-program), The Centre for Disease Control for the Northern Territory (https:// health.nt.gov.au/professionals/centre-for-diseasecontrol/cdc-contacts); the Mosquito Management and Asset Services in the Brisbane City Council for Queensland (https://www.brisbane.gld.gov.au/ clean-and-green/natural-environment-and-water/ biodiversity-in-brisbane/wildlife-in-brisbane/ invasive-plants-and-animals/mosquitoes); the Department of Primary Industries for Victoria (https://www.betterhealth.vic.gov.au/health/ healthyliving/mosquito-management-victoria); and the University of South Australia for South Australia (https://www.sahealth.sa.gov.au/wps/ wcm/connect/public+content/sa+health+internet/ public+health/pest+management/mosquitoes/ arbovirus+and+mosquito+monitoring+reports & Email: mozzie.monitors@unisa.edu.au). Human Population Data: Australian Bureau of Statistics (https://dbr.abs.gov.au/).

**Funding:** Financial support for this project was provided through the Funding Initiative for Mosquito Management In Western Australia Research Grants (to SC and ISK), administered by the Department of Health, Western Australia. The funders had no role in study design, data collection

#### Author summary

Ross River virus (RRV) causes the highest number of vector-borne disease infections in Australia, yet the mechanisms driving its transmission across regions remains poorly understood. We analyzed long-term surveillance data from eight epidemic regions spanning tropical to temperate climates. We tested the importance of different mosquito vectors, wildlife hosts, and varies seasonal and transmission effects in explaining observed patterns of RRV notifications. Despite differing environments, models indicate combinations of two key mosquito vectors and two marsupial hosts, and interacting seasonally best explains RRV dynamics. Estimated mosquito-human transmission rates were similar across regions, whereas wildlife contributions varied. Models estimate 43% underreporting of RRV infections nationally. Findings provide new quantitative insights on transmission mechanisms and health impacts of RRV. Estimating mechanisms and key parameters allows for the future assessment of public health interventions like mosquito control. This modelling framework evaluating long-term data could be applied to other complex vector-borne diseases to unravel intrinsic drivers and guide mitigation strategies.

## Introduction

Spillover into humans from non-human animal sources is a common characteristic of many vector-borne pathogens, including those classified as emerging or resurging. Despite the significant public health burden of vector-borne pathogens, the specific intrinsic mechanisms driving spillover and outbreak events (beyond increases in vector abundance) are however often poorly understood. Fundamentally, this is because the epidemiological dynamics of vector-borne diseases are complex, including a web of extrinsic (environmental and meteorological) and intrinsic (vector, host, and pathogen biology) factors, as well as features from public health (e.g., vector and host control, diagnostic standards, public behaviour, and education). This complexity presents a significant challenge to understanding the combinations of mechanisms most responsible for observed disease dynamics, yet a deeper understanding of the intrinsic mechanisms underscoring human incidence of vector-borne zoonoses is important to guide and improve public health management.

Ross River virus (RRV; family: Togaviridae, genus: Alphavirus) has the highest incidence of any vector-borne disease in Australia and is a classic illustration of complex mechanisms driving diverse disease dynamics [1-4]. The annual notification incidence rate for RRV is > 40 per 100,000 population, with an estimated annual health care and lost productivity cost of \$15 million [1,5]. There are 42 known mosquito species (spanning seven genera) capable of transmitting RRV, with the primary vector mosquito species responsible for human transmission being Aedes vigilax, Culex annulirostris, Aedes camptorhynchus, and Aedes notoscriptus [6]. Several key reservoir hosts contribute to RRV transmission, all of which vary in relative competence and abundance across epidemic regions [4,7,8]. Marsupials (e.g., kangaroos, wallabies, & possums) are generally considered the most competent and important host reservoirs of RRV. However, placental mammals and birds may also contribute to transmission, adding to the ecological complexity in mechanisms driving disease occurrence [4]. RRV is sustained enzootically between mosquitoes and non-human mammals, and evidence indicates that epizootics lead to spillover and human epidemics [3,9-11]. Transmission dynamics of RRV are also variable across geographic regions, often linked with the effects of climate on mosquito species compositions and abundance [2,8,12,13]. Changes in climate can drive shifts in species community compositions and the dominance of specific species. Clinical symptoms of RRV in and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

humans include fever, polyarthralgia, rashes, polyarthritis, lymphadenopathy, lethargy, headaches, myalgias, photophobia, and glomerulonephritis [1,5,6]. The duration and extent to which symptoms persist vary between individuals, however, a typical period of morbidity extends from 3–6 months but, in some instances, can exceed one year [5,6].

This study combines diverse and extensive data to advance understanding of the intrinsic mechanisms that underpin mosquito-borne viral transmission associated with spillover and human incidence. We focus on RRV, owing to the exceptionally high-quality national surveillance of human notifications and mosquito vector surveillance spanning multiple decades, across multiple epidemic centres in Australia. This rich epidemiological surveillance of long-term human RRV incidence has been crucial in establishing valuable predictive disease surveillance systems for outbreak detections, particularly using environmental and vector populations to aid in public health management. However, appreciation of which intrinsic factors are most important in epidemic centres has remained elusive (e.g., the most important vector species, host species and other mechanisms governing human incidence at a site and across sites). Furthermore, advancing mechanistic understanding of human RRV incidence is essential because climatic effects and vector abundance do not always predict human outbreaks [1,13–16].

Beyond vector and host species abundance, a range of other intrinsic mechanisms can also impact the dynamics of RRV. For example, waning host immunity may be key for epizootics, spillover, and epidemics to occur [7]. This may explain why vector abundance is not always a reliable indicator of epidemics [17,18]. Seasonal variation in vector feeding rates (associated with temperature) or host preferences also has the potential to influence seasonality in transmission. Vertical transmission in vectors can also initiate and sustain local RRV transmission. Reservoir hosts may also intermittently shed the virus if the virus is able to occasionally escape immune control (recrudescence), such as when the host is under stressful conditions that result in immune suppression [19]. The relative importance of these potential factors may also vary geographically, such as among tropical northern and temperate southern epidemic centres in Australia. Collectively, there is a need for a deeper understanding of the relative importance of varying intrinsic mechanisms that drive observed dynamics of RRV in Australia-a problem common to other vector-borne zoonoses of public health importance globally. By combining the long-term surveillance of human incidence of vector-borne disease and vector populations with information on reservoir host communities and other intrinsic factors, research can begin to test potential mechanisms driving variation in the pattern of human incidence and contrast these epidemiological systems across epidemic centres.

In this study, we bring together, to the best of our knowledge, the most extensive and diverse data on a vector-borne disease in Australia to assess the relative potential of hypothesised transmission mechanisms influencing the dynamics of RRV incidence. We utilise longterm empirical data on human and vector populations and integrate these, with other vector/ host/pathogen parameters, into multi-vector, multi-host, multi-transmission mechanistic Susceptible-Infected-Recovered (SIR) models. We fit our models against high-quality long-term RRV surveillance data from humans, estimate a range of critical parameters (e.g., transmission rates), and use model selection to evaluate the importance of varying mechanisms involved in disease transmission. This study focuses on estimating parameters associated with RRV transmission dynamics and evaluating model fits to observed data retrospectively. Our aim is not to develop a prospective forecasting tool per se, but rather to use the extensive long-term data to gain ecological insights into transmission mechanisms. While forecast modelling approaches are important for prediction, here we take advantage of the rich dataset to look retrospectively and build mechanistic understanding of the drivers and parameters underlying observed dynamics. The primary aims of this study are to (1) evaluate the relative likelihood of hypothesised mechanisms responsible for observed human RRV dynamics; (2) estimate key

parameters associated with the transmission of RRV and reporting of RRV notifications; and (3) assess how hypothesised mechanisms driving RRV dynamics vary, or are similar, among epidemic sites around Australia.

## Methods

#### Model structure

In this study, we use deterministic Susceptible-Infected-Recovered ordinary differential equation (ODE) models [20] to investigate the ecological mechanisms governing RRV for each study site. There were eleven ecological scenarios fitted to the RRV notification data using a variety of vector, host, and transmission parameters to investigate mechanisms driving RRV transmission (Please refer to the Model selection section below). Our full model includes up to two vector species (see Vector Monitoring and Competence Data below), three host species (see Reservoir Host Population and Competence Data below), and multiple mechanisms to potentially shape transmission (System of Eq 1). Accordingly, the transmission of RRV in our model consists of up to five populations: three host species and two mosquito species, with mosquito vectors divided into two subgroups of susceptible (S), infected (I), and hosts divided into three subgroups susceptible, infected, and recovered (R) classes. The rate of RRV transmission from mosquitoes to hosts is assumed to be frequency-dependent, whereby the rate of transmission increases with the total proportion of the vector population which is infected and therefore allowing for disease transmission to persist even with low host densities [7,21]. The rate of RRV transmission from hosts to mosquitoes is assumed to be density-dependent, whereby the contact rate between susceptible and infected mosquitoes with hosts depends upon both population densities, with higher densities increasing transmission rates [22].

N = S + I

The full model is described mathematically as follows

$$\begin{split} V_{v_{1}} &= v_{v_{1}} + v_{v_{1}} \\ &N_{v_{2}} = S_{v_{2}} + I_{v_{2}} \\ &N_{k} = S_{k} + I_{k} + R_{k} \\ &N_{p} = S_{p} + I_{p} + R_{p} \\ &N_{h} = S_{h} + I_{h} + R_{h} \\ \\ &\frac{dS_{v_{1}}}{dt} = b(t)_{v_{1}}(1 - \phi_{v_{1}})N_{v_{1}} - \beta_{g} \left(\beta_{kv_{1}} \left(\frac{S_{v_{1}}I_{k}}{N_{v_{1}}}\right) - \beta_{pv_{1}} \left(\frac{S_{v_{1}}I_{p}}{N_{v_{1}}}\right) - \beta_{hv_{1}} \left(\frac{S_{v_{1}}I_{h}}{N_{v_{1}}}\right)\right) - d_{v_{1}}S_{v_{1}} \\ &\frac{dI_{v_{1}}}{dt} = b(t)_{v_{1}}\phi_{v_{1}}N_{v_{1}} + \beta_{g} \left(\beta_{kv_{1}} \left(\frac{S_{v_{1}}I_{k}}{N_{v_{1}}}\right) + \beta_{pv_{1}} \left(\frac{S_{v_{1}}I_{p}}{N_{v_{1}}}\right) + \beta_{hv_{1}} \left(\frac{S_{v_{1}}I_{h}}{N_{v_{1}}}\right)\right) - d_{v_{1}}I_{v_{1}} \\ &\frac{dS_{v_{2}}}{dt} = b(t)_{v_{2}}(1 - \phi_{v_{2}})N_{v_{2}} - \beta_{g} \left(\beta_{kv_{2}} \left(\frac{S_{v_{2}}I_{k}}{N_{v_{2}}}\right) - \beta_{pv_{2}} \left(\frac{S_{v_{2}}I_{p}}{N_{v_{2}}}\right) - \beta_{hv_{2}} \left(\frac{S_{v_{2}}I_{h}}{N_{v_{2}}}\right)\right) - d_{v_{2}}S_{v_{2}} \end{split}$$

 $\frac{dI_{v_2}}{dt}$ 

$$\begin{split} = b(t)_{v_{2}}\phi_{v_{2}}N_{v_{2}} + \beta_{\theta}\left(\beta_{kv_{2}}\left(\frac{S_{v_{2}}I_{k}}{N_{v_{2}}}\right) + \beta_{pv_{2}}\left(\frac{S_{v_{2}}I_{p}}{N_{v_{2}}}\right) + \beta_{hv_{2}}\left(\frac{S_{v_{2}}I_{h}}{N_{v_{2}}}\right)\right) - d_{v_{2}}I_{v_{2}} \\ \\ \frac{dS_{k}}{dt} = b(t)_{\theta k}\left(N_{k} - d_{k}S_{k} - \beta_{v_{1}k}S_{k}I_{v_{1}} - \beta_{v_{2}k}S_{k}I_{v_{2}}\right) \\ \frac{dI_{k}}{dt} = \rho(t)_{\theta k}\left(S_{k}\left(\beta_{v_{1}k}\frac{I_{v_{1}}}{N_{v_{1}}} + \beta_{v_{2}k}\frac{I_{v_{2}}}{N_{v_{2}}}\right)\right) - (\gamma_{k} + d_{k})I_{k} + \omega_{k}R_{k} \\ \\ \frac{dR_{k}}{dt} = \gamma_{k}I_{k} - (d_{k} + \omega_{k})R_{k} \\ \\ \frac{dS_{p}}{dt} = b(t)_{\theta p}\left(N_{p} - d_{p}S_{p} - \beta_{vp}S_{p}I_{v_{1}} - \beta_{v_{2}p}S_{p}I_{v_{2}}\right) \\ \\ \frac{dI_{p}}{dt} = \rho(t)_{\theta p}\left(S_{p}\left(\beta_{v_{1}p}\frac{I_{v_{1}}}{N_{v_{1}}} + \beta_{v_{2}p}\frac{I_{v_{2}}}{N_{v_{2}}}\right)\right) - (\gamma_{p} + d_{p})I_{p} + \omega_{p}R_{p} \\ \\ \\ \\ \frac{dS_{h}}{dt} = b(t)_{h}N_{h} - d_{h}S_{h} - \beta_{v_{1}h}S_{h}I_{v_{1}} - \beta_{v_{2}h}S_{h}I_{v_{2}} \\ \\ \\ \frac{dI_{h}}{dt} = \rho(t)_{\theta h}\left(S_{h}\left(\beta_{v_{1}h}\frac{I_{v_{1}}}{N_{v_{1}}} + \beta_{v_{2}h}\frac{I_{v_{2}}}{N_{v_{2}}}\right)\right) - (\gamma_{h} + d_{h})I_{h} \\ \\ \\ \\ \\ \\ \frac{dR_{h}}{dt} = \gamma_{h}I_{h} - d_{h}R_{h} \end{split}$$
(1)

In the above, *t* is time, *S<sub>i</sub>* is the number of animals within species *i* that are susceptible; *I<sub>i</sub>* the number that are infected; *R<sub>i</sub>* the number that are recovered. Host and vector species *i* are defined as follows: *k* for kangaroos/wallabies; *p* for possums; *h* for humans; *v*<sub>1</sub> for primary vector; and *v*<sub>2</sub> for secondary vector, respectively. The seasonally varying birth rate of species *i* is represented by  $b(t)_{i}$ ;  $\phi_i$  the vertical transmission rate; *d<sub>i</sub>* the mortality rate;  $\beta_{vi}$  the transmission rate from vector *v*<sub>1</sub> to host species *i*;  $\rho(t)_{\vartheta_i}$  the seasonally varying disease transmission rate to host species *i*;  $\gamma_i$  the recovery rate of species *i*; *N<sub>i</sub>* the population size; and  $\omega_i$  the number of species *i* that are recrudescent.

Our transmission models consider seasonal phases and amplification effects of vector feeding on RRV transmission to host species *i* expressed in Eq 2, denoted by  $\rho(t)_{\vartheta i}$ . In Eq 2, two seasonal parameters are estimated, the phase, denoted by  $\epsilon_{\mu i}$ , and amplitude of seasonality, denoted by  $\sigma_{\varphi i}$ , of RRV transmission and used in the transmission model in Eq 1. For regions and species where seasonal variation was not relevant (see Table 1),  $\sigma_{\varphi i}$  was fixed to zero.

$$\rho(t)_{\vartheta i} = 1 + \sigma_{\varphi i} \sin(2\pi(t - \epsilon_{\mu i})) \tag{2}$$

Parameter	Definition	Darwin (NT)	Brisbane (Qld)	Mandurah (WA)	Mildura (Vic)	Gippsland (Vic)	Renmark (SA)	Murray Bridge (SA)	Coorong (SA)
N <sub>k</sub>	Macropod density (per km <sup>2</sup> )	8	4.7	21.9	1.91	10.7	4.52	4.52	4.52
$N_p$	Possum density (per km <sup>2</sup> )	300	34.7	5.7	100.6	19.6	11.1	12.1	1.0
$\overline{\theta_k}$	Phase between macropod births (proportion of years)	0.25 (0.22–0.29)	0.48 (0.42–0.51)	0.37 (0.32–0.41)	0.54 (0.51-0.58)	0.30 (0.27–0.33)	0.54 (0.51–0.57)	0.23 (0.20-0.26)	0.09 (0.06-0.12)
$\overline{ heta_p}$	Phase between possum births (proportion of years)	0.81 (0.62–1.00)	0.13 (0.05–0.20)	0.51 (0.42–0.58)	0.43 (0.37–0.49)	0.31 (0.25–0.37)	0.43 (0.37–0.49)	0.24 (0.18–0.30)	0.11 (0.05-0.17)
$\overline{\delta_k}$	Seasonal reproduction	No	Yes	Yes	Yes	Yes	Yes	Yes	Yes
$\overline{\delta_p}$	Seasonal reproduction	No	Yes	Yes	Yes	Yes	Yes	Yes	Yes
k	Macropod species	AW	EGK	WGK	EGK	EGK	RK	RK	RK
p	Possum species	BTP	BTP						
$\overline{\nu_1}$	Primary vector species	AeV	AeV	AeC	CuA	AeC	AeC	AeC	AeC
<i>v</i> <sub>2</sub>	Secondary vector species	CuA	CuA	AeV	AeC	CuG	CuA	CuA	CuG
Site	Area size (km <sup>2</sup> )	92	1343	2948	512	2229	916	1832	8833
characteristics	Number of households	-	16661	-	18393	15409	3627	7935	2107

Table 1. Site-specific invariant parameters of host and vector species, seasonal birth rates (range), and site characteristics.

https://doi.org/10.1371/journal.ppat.1011944.t001

It is noted that human epidemics of RRV are not always preceded by high vector abundance and that these epidemics may be driven by a decline in seroprevalence within host populations and the recruitment of new non-immune hosts, amplifying transmission [11,23–25]. Seasonally forced birth rates were used to account for seasonal vector feeding and the timing of recruitment of susceptible newborn hosts, seen in Eq.3. The relative phase for species *i* was estimated, denoted by  $\theta_i$ , by using nonlinear least-squares to fit the proportion of the host population that breed per month for each host and lagged to adjust for the time it takes for new offspring to leave the pouch and become accessible to feeding vectors. This was modelled by a sinusoidal function of the proportion of months within a year (Eq.3). Phase between host breeding is represented as a proportion in years, where susceptible host offspring are exposed to feeding vectors. Additionally, vector and host populations are assumed to be stable and unlikely to become extinct over time.

$$b(t)_{v_1} = (1 - \delta_{v_1} \sin(2\pi t)) d_{v_1}$$
  

$$b(t)_{v_2} = (1 - \delta_{v_2} \sin(2\pi t)) d_{v_2}$$
  

$$b(t)_k = (1 - \delta_k \sin(2\pi (t - \theta_k))) d_k$$
  

$$b(t)_p = (1 - \delta_p \sin(2\pi (t - \theta_p))) d_p$$
  

$$b(t)_h = d_h$$
  
(3)

#### Study sites

RRV transmission was modelled across eight sites from five States and Territories in Australia, selected based on their high mosquito-borne disease attack rates, mosquito surveillance programs, spatial and epidemiological differences, and epidemiological importance [1,6– 8,17,23,26,27]. The eight study sites span tropical, subtropical, and temperate climates and are likely to have different mosquito species compositions and dynamics driving local RRV transmission. Sites include Mandurah (Western Australia), Darwin (Northern Territory), Brisbane



Fig 1. Map location of study sites, showing Darwin from a tropical region, Brisbane from a sub-tropical region, and Mandurah, Mildura, Gippsland, Renmark, Murray Bridge, and Coorong from a Mediterranean region. The basemap data were collected from the Australian Bureau of Statistics ('States and Territories—2021–Shapefile', https://www.abs.gov.au/statistics/standards/australian-statistical-geography-standard-asgs-edition-3/jul2021-jun2026/access-and-downloads/digital-boundary-files).

https://doi.org/10.1371/journal.ppat.1011944.g001

(Queensland), Mildura and Gippsland (Victoria), and Coorong, Murray Bridge, and Renmark (South Australia) (Fig 1). Each site comprised different study areas, as follows: in Mandurah, the Local Government Areas (LGA) of Rockingham, Mandurah, Murray, and Waroona; in Darwin, the Statistical Areas (SLA) level 2 of Alawa, Anula, Brinkin-Nakara, Coconut Grove, Jingili, Karama, Leanyer, Lyons, Malak, Marrara, Millner, Moil, Nightcliff, Rapid Creek, Tiwi, Wagaman, Wanguri, and Wulagi; in Brisbane, the Brisbane LGA; in Mildura, the SLA level 2 areas of Mildura (North and South), Red Cliffs, Irymple, and Merbein; in Gippsland, the SLA level 2 areas of Wellington, Sale, Longford, Paynesville, and Bairnsdale; in Renmark, the Renmark Paringa LGA; in Murray Bridge, the Murray Bridge LGA; and in Coorong, the Coorong LGA. These LGAs and SLAs are defined by the Australian Bureau of Statistics and make up the catchment areas where mosquito monitoring programs are actively carried out. These areas were selected because they capture the human populations in proximity with the mosquito populations being monitored. All data used in the following sites and models were collected between 1991–2014 for Mandurah, Darwin and Mildura, 1998–2013 for Brisbane, 1996–2013 for Gippsland, and 1999–2017 for Renmark, Murray Bridge, and Coorong.

Common names, genus and species and reference, hosts include AW, Agile wallaby (*Macropus agilis*); EGK, Eastern grey kangaroo (*Macropus giganteus*); WGK, Western grey kangaroo (*Macropus fuliginosus*); RK, red kangaroo (*Macropus rufus*); BTP, Brush-tailed possum (*Trichosurus vulpecula*); and vector include: AeV, *Aedes vigilax*; CuA, *Culex annulirostris*; AeC, *Aedes camptorhynchus*; and CuG, *Culex globocoxitus*. Host species and density estimates (and confidence intervals) were sourced from the literature as follows: Darwin [28–32]; Brisbane [8,32–34]; Mandurah [8,24,31,32]; Mildura [32,35–38]; Gippsland [32,35–37]; and Renmark, Murray Bridge, and Coorong [8,32,39–41].

#### Human incidence and population data

Data for de-identified RRV human notifications per week were provided for each of our study sites by State and Territory from the following sources: in Western Australia, Department of Health and the Western Australian Notifiable Infectious Diseases Database; in Northern Territory, Northern Territory Department of Health and Centre of Disease Control Surveillance Unit; in Queensland, Queensland Department of Health and the Communicable Disease Branch; in Victoria, Victorian Department of Health and the Communicable Disease Epidemiology and Surveillance Branch; and in South Australia, Department of Health South Australia and the Communicable Disease Control Branch. Annual population data were collected from historical and current records from the Australian Bureau of Statistics for each study site [42]. Annual human population data were interpolated, using piecewise linear interpolation, into weekly observations within each year. Using human population data collected from the Australian Bureau of Statistics [32], we calculated RRV notifications per 100,000 individuals per week, which was used to fit the deterministic models.

RRV human notification data were combined with mosquito surveillance data by week and year (vector monitoring data is detailed below). Where mosquito monitoring was unavailable (ranging from 0% to 78% missing) for weeks with RRV notifications, mosquito species densities interpolated using cubic spline interpolation. Our data also retained observations where mosquito monitoring had been conducted, but no RRV notifications were observed.

#### Vector monitoring and competence data

Weekly mosquito trapping count data were provided by each respective State and Territory's mosquito surveillance and control programs, including The Arbovirus Surveillance and Research Laboratory for Western Australia, The Centre for Disease Control for the Northern Territory; the Mosquito Management and Asset Services in the Brisbane City Council for Queensland; the Department of Primary Industries for Victoria; and the University of South Australia for South Australia. Two vector species were used in the following models based on their known relative importance in RRV transmission and relative abundance in each site.

There are four major mosquito vector species responsible for the majority of RRV transmission: *Aedes vigilax, Culex annulirostris, Aedes camptorhynchus*, and *Aedes notoscriptus*. These species also typically make up the greatest proportion of mosquito community abundance in epidemic centres, supporting our preliminary analysis. We restricted our analysis to two species of vector mosquito to prevent overfitting our models, as at each site, the other species of mosquito were at very low densities in comparison (See <u>S1 Appendix</u> for species abundance over time). The relative abundance and dynamics of these major RRV vector species can vary across regions based on climate and habitat. For example, *Ae. vigilax, Cu. annulirostris*, and

Population	pulation Parameter Rate Definition (frequency)		Value (Range)	Source		
<i>Vector</i> $\phi_{v_i}$ Vertical transmission & larval survival (daily)		Vertical transmission & larval survival (daily)	8.5 x 10 <sup>-4</sup>	[35]		
	$d_{v_i}$	Mortality rate (daily)	0.10 (0.02–0.33)	[43]		
Human	$\gamma_h$	Recovery rate (annual)	$\frac{365.25}{4(1-6)}$	[34,53,54]		
Macropods	$\gamma_k$	Recovery rate (annual)	$\frac{365.25}{6(2-6)}$	[53]		
	$d_k$	Mortality rate (annual)	$log(1+b(t)_k)$	A relatively stable population over time		
Possums	$\gamma_P$	Recovery rate (annual)	$\frac{365.25}{1(1-3)}$	[55,56]		
	$d_p$	Mortality rate (annual)	$log(1+b(t)_p)$	A relatively stable population over time		

Table 2. Invariant population parameters shared across sites of host and vector species recovery rate of infection, general mortality rates, and vertical transmission of RRV in vector populations.

https://doi.org/10.1371/journal.ppat.1011944.t002

*Ae. camptorhynchus* differ in host feeding preferences and the timing of population peaks. We selected the most relatively abundant species at each site as likely important to local transmission patterns. The selection of only two vector mosquitoes in these models does not reflect that other mosquito species do not contribute to RRV transmission. This choice was top balance model complexity and the low populations of other vector species.

Each site had several mosquito trapping stations used to capture mosquitoes, and we calculated the mean weekly number of mosquito species per trap. The mosquito species used for each site were then selected based on their relative abundance within the mosquito species community. The mosquito species with the greatest maximum abundance was classified as the primary vector and the second most abundant as the secondary vector. The use of the term "secondary vector" does not indicate that this vector is any more of less important for transmission than the primary vectors, it merely represents a second vectors species in the transmission system being investigated. It is assumed that these mosquito species populations do not become extinct but have a constant mortality rate of 10% per day (Table 2) [41,43–49]. The primary and secondary vectors for each site can be found in Table 1.

Our models estimate the maximum and minimum mosquito population density per week, denoted by  $v_{1max}$  and  $v_{1min}$ , respectively, which populations could not exceed or fall below. Furthermore, the relative abundance of the secondary mosquito population, denoted by  $v_2$ , compared with the primary vector species was also estimated. The parameter  $v_{1min}$  was constrained such that it could not exceed 1% of the maximum of primary vector species based on raw mosquito trap data, which is an average across all the traps at each site. Furthermore, secondary mosquito vector populations are scaled to that of the maximum observed primary vector population size in the models presented here. Vertical transmission in vector mosquitoes varies, having different effects on the epidemiology of diseases by location influencing disease persistence, transmission, and magnitude of epidemics [35,50–52]. We consider RRV vertical transmission, denoted as  $\phi_{v_i}$  (Table 2), among our vectors of 0.85 per 1,000 mosquitoes using known estimates from *Aedes camptorhynchus* [35] and generalised to the other mosquito species investigated here. Empirical mosquito population data were used to simulate vector abundance through time, allowing for parametrisation of an estimated birth rate for our vector species (Eq 3).

#### Reservoir host population and competence data

Host species information used in the following models is detailed in Table 1. Host population data are often limited, giving a single snapshot of a population's abundance with no information regarding temporal change [8]. Therefore, we use a point estimate for animal host abundance when parametrising our models (Table 1). Given the paucity of host population data

available and the relative complexity around determining the potential host species involved in RRV transmission [2-4], we focused on only two host reservoirs in our transmission models. Macropods and possums were selected owing to their known competence and relatively high abundance in the epidemic centres investigated here [8]. Whilst there are other ecologically important hosts that also contribute to RRV transmission (i.e., birds) and add to the ecological complexity in mechanisms driving disease occurrence, we used marsupial hosts owing to their known role in RRV transmission and our ability to reasonably approximate their populations. Where possible, species densities are taken from peer-reviewed literature and converted into a density representing the number of individual species per km<sup>2</sup>. For species where exact densities were unavailable (such as with possums), we made approximations using methods developed in Koolhof & Carver (2017) based on the number of households within each site to then derive species densities per km<sup>2</sup>. Host mortality rates vary greatly depending on multiple seasonal, environmental, and climatic differences. Owing to this variability and lack of continuous and reliable mortality estimates, we assume relatively stable host populations through time (Table 2). It is assumed that the recovery rate of an RRV infection to be the average number of days in the year (365.25) divided by the viremic duration in days of host species i (Table 2). The host viremic periods used here are based on empirically founded estimates and do not account for variation in length of viremia, nor do we account for viral titres.

Monthly host breeding estimates (percent of the population breeding) were collected from the literature closest to our study sites in Darwin [31,57], Brisbane [58], with macropod information coming from Victorian estimates [59,60], Mandurah [61–64], and Mildura, Gippsland, Renmark, Murray Bridge, and Coorong [37,59,60,65]. For Darwin, marsupial reproduction is typically seen year-round, reflective of the tropical climate, and thus was not considered to have seasonal breeding (Table 1). Breeding estimates were adjusted by the average time between conception and being 'young at foot' (out of the pouch and exposed to feeding vectors) for macropods, and the time it takes for possums to become back riders; 7–9 and 5 months, respectively. The relative amplitudes of the seasonally forced birth rates are denoted by  $\delta_i$  for species *i* (Eq 3). Moreover, the timing of host reproduction varies latitudinally in Australia, with the reproduction and breeding of hosts in southern latitudes being seasonally driven when compared with hosts further north [59]. These models include the forcing of annual seasonal breeding in our southern latitudes, compared with the potential for continuous and less seasonally driven breeding in northern latitudes.

#### Model fitting and parameter estimation

Our models fit human RRV incidence data (see Human incidence and population data below) at a weekly level collected at State and Federal levels as part of routine surveillance of nationally notifiable diseases. Parameters were estimated assuming a beta-binomial distribution allowing for overdispersion, and model section on maximum likelihood estimations. RRV has long been regarded to be considerably underreported/undiagnosed, with wide variation in the severity of symptoms RRV infections cause, asymptomatic infections, and individual behavioural differences in presentations to general practitioners [1,12,66,67]. We account for this variation in reporting by estimating a false-negative reporting rate, more commonly referred to as a notification fraction [68]. Furthermore, as changes in the RRV national case definition in 2006 led to increased false-positive notifications of RRV infections by including non-incident cases, we also estimated a false-positive reporting rate, from 2006 onwards, within our models using a Heaviside step function [69], denoted H(t). Within the beta-binomial, we estimate three additional parameters; a false negative rate denoted by  $\beta_r$ , a false positive reporting rate of RRV notifications denoted by  $\alpha_r$ , an overdispersion parameter, denoted by s, to account

for variation in reporting dynamics through time and the variation in RRV prevalence (Eq 4). The beta-binomial probability mass function is given in Eq.4, where f is the probability density of the beta-binomial distribution,  $\phi$  is the set of model parameters (including s),  $\varepsilon$  is the number of observed RRV notifications at each reported time point, n is the observed human population size at each time point, m is the number of modelled notifications (which depends on the model parameters  $\phi$ ) at each time point, and s is the overdispersion parameter.

$$m(t) = (1 - \beta_r)I_h + \alpha_r H(t)$$

$$H(t) = \begin{cases} 1 & \text{if } t \ge 2006 \\ 0 & \text{otherwise} \end{cases}$$

$$\alpha_r = sm \text{ and } \psi = s(1 - m) \qquad (4)$$

$$f(\varepsilon|n, m, s) = \binom{n}{\varepsilon} \frac{B(\varepsilon + \alpha, n - \varepsilon + \psi)}{B(\alpha, \psi)}$$

$$max L(\phi|\varepsilon) = max f(\varepsilon|n, m, s)$$

-- ( )

#### Model selection

Model combinations were selected based on biologically plausible transmission pathways. Eleven ecological scenarios were fitted to the RRV notification data (Table 3) using combinations of multiple mosquitoes and host species and seasonal vector feeding parameters in deterministic ODE models (Eq 1). Development of these model combinations started with the simplest transmission pathway (e.g., assuming there to be only a single primary vector and human population), then adding increasing complexity which could plausibly contribute to RRV transmission, by considering additional host and vector populations, host recrudescence, and transmission seasonality. Some parameters included in the models could only be included if others were also included. For instance, host recrudescence could only be included if the transmission between vector and host *i* was also included in the model combination (Table 3). Maximum likelihoods were estimated assuming a beta-binomial distribution using 'lsode', a statistical solver for ordinary differential equations, that utilised a general-purpose optimization method based on Nelder-Mead. Akaike's Information Criterion (AIC) values for each model combination were derived from the maximum likelihood estimates to find and rank the models using delta ( $\Delta$ ) AIC [70]. Parameter estimates were then averaged over all model combinations using model weights derived from the  $\Delta$ AIC values [71].

To assess the capacity of the mechanistic models to qualitatively capture the dynamics of human RRV incidence, we compared the observed number of outbreaks to the model fitted the number of outbreaks using positive and negative predicted values (PPV and NPV respectively). As our modelling is not for predictive or forecasting purposes, to avoid confusion, we have re-defined these terms here as positive validated outbreak (PVO) and negative validated outbreak (NVO). Here we classified an outbreak of RRV to occur if the number of notifications was above the mean number of RRV notifications per 100,000 plus one standard deviation, calculated over the entire time period for each site [15]. The outbreak analysis using PVO and NVO was intended to provide a qualitative view of whether models could generally

Scenario	Transmission model description	Parameters estimated
1	Primary vector with humans	$\beta_{vh}, v_{1min}, v_{1max}$
2	Primary vector with humans and kangaroos	$\beta_{vh}, \beta_{vk}, v_{1min}, v_{1max}$
3	Primary and secondary vector with humans, and kangaroos	$\beta_{vh}, \beta_{vk}, v_{1min}, v_{1mx}, v_2$
4	Primary vector with humans, and kangaroos. Seasonality in vector feeding (transmission)	$\beta_{vh}, \beta_{vk}, v_{1min}, v_{1max}, \rho_{\vartheta}$
5	Primary vector with humans, kangaroos, and possums.	$\beta_{vh}, \beta_{vk}, \beta_{vp}, v_{1min}, v_{1max}$
6	Primary and secondary vector with humans, kangaroos, and possums	$\beta_{vh}, \beta_{vk}, \beta_{vp}, v_{1min}, v_{1max}, v_2$
7	Primary vector with humans, kangaroos, and possums. Seasonality in vector feeding (transmission)	$\beta_{vh}, \beta_{vk}, \beta_{vp}, v_{1min}, v_{1max}, \rho_{s}$
8	Primary and secondary vector with humans, kangaroos, and possums. Seasonality in vector feeding (transmission)	$\beta_{vh}, \beta_{vk}, \beta_{vp}, v_{1min}, v_{1max}, v_{2}$ $\rho_{\vartheta}$
9	Primary vector with humans, kangaroos, and possums. Recrudescence in the kangaroo and possum	$ \begin{array}{c} \beta_{vh}, \beta_{vk}, \beta_{vp}, v_{1min}, v_{1max}, \\ \omega_k, \omega_p \end{array} $
10	Primary vector with humans, kangaroos, and possums. Seasonality in vector feeding (transmission). Recrudescence in the kangaroo and possum	$ \begin{array}{c} \beta_{\nu h}, \beta_{\nu k}, \beta_{\nu p}, \nu_{1 min}, \nu_{1 max}, \\ \rho_{\vartheta}, \omega_k, \omega_p \end{array} $
11	Primary and secondary vector with humans, kangaroos and possums. Recrudescence in the kangaroo and possum. Seasonality in vector feeding (transmission)	$\beta_{vh}, \beta_{vk}, \beta_{vp}, v_{1min}, v_{1max}, v_{2}$ $\rho_{\vartheta}, \omega_{k}, \omega_{p}$

Table 3. Transmission scenarios model combinations for explaining RRV transmission across epidemic centres in Australia.

Where,  $\beta_{vi}$  the transmission rate from vectors to host species *i*;  $\omega_i$  rate of recrudescence in host species *i*;  $\epsilon_{\mu}$  the seasonal phase of transmission;  $\sigma_{\varphi}$  the amplitude of seasonal transmission;  $v_{1min}$  minimum mean estimated vector population in proportion to the maximum observed vector population per mosquito monitoring trap per week;  $v_{1max}$  estimated maximum mean vector population per mosquito monitoring trap per week; and  $v_2$  estimated mean secondary vector population per mosquito monitoring trap per week. Host and vector species *i* are defined as kangaroos/wallabies, *k*; possums, *p*; and humans, *h*. In each scenario, "X" indicates if a transmission factor was included in the model combination.

https://doi.org/10.1371/journal.ppat.1011944.t003

capture observed epidemic patterns. It was not used as a quantitative validation metric to assess model fit. Model selection and parameterization were based on maximum likelihood estimation and AIC comparisons of different model combinations. Analyses were performed in R v3.5.3 (R Core Team 2018) using the packages 'lubridate' [72], 'rmutil', 'caret', and 'deSo-lve' [73] in RStudio (Version 1.2.1335). The ODE solver and R code can be found in the supplementary material (S2 Appendix).

#### Results

RRV incidence in Darwin, Brisbane, and Mandurah is near year-round with annual and/or near biennial epidemic seasons (Figs 2 and 3). These areas typically have a build-up of RRV infections over several weeks which characterises epidemic periods, potentially suggesting sustained transmission and spillover from reservoir hosts to human populations. In contrast to these more northern sites, RRV transmission is seasonal in the southern temperate sites of Mildura, Gippsland, Coorong, Murray Bridge, and Renmark, and it is common to have several years between RRV epidemic seasons, with notable periods of no RRV notifications in humans (Figs 2 and 3). Epidemics in these regions occur abruptly with epidemic transmission being short-lived and generally with little reported incidence between epidemics.

We modelled RRV transmission using parameter combinations of vector, host, and transmission (seasonality in host birth and vector feeding, and recrudescence (previously recovered hosts intermittently shedding the virus) in marsupials; see Table A in <u>S3 Appendix</u> for model combinations). Of these models, the most likely (i.e., models with  $\Delta AIC < 4$ ) fitted to RRV



**Fig 2.** Per capita (100,000 people) RRV notifications over time for Darwin, Brisbane, Mandurah, and Mildura showing the recorded data (grey) and weighted model averages from the models (red) (<u>Table 3</u>).

https://doi.org/10.1371/journal.ppat.1011944.g002

notifications were typical of moderate to high complexity (Table 4). Notably, all most likely models across sites included the primary and secondary vector species and transmission between macropods and possums. Seasonality in vector feeding as a parameter was included among the most likely model for each site, except Mandurah (Table 4). Despite the complexity





https://doi.org/10.1371/journal.ppat.1011944.g003

and humans, h	Best fitting model	ls ( $\Delta AIC < 4$ ) are i	n bold, with the mo	st likely model fit	having a $\Delta AIC = 0$ .	species <i>i</i> are defined	a as kaligaroos/ wallables,	, k; possuiiis, p;
Scenario	Darwin	Brisbane	Mandurah	Mildura	Gippsland	Renmark	Murray Bridge	Coorong
1	1466.6	58.3	981.9	90.3	1276.6	194.8	270.5	352.5
2	1467.9	55.3	983.9	93.3	413.2	202.0	184.7	354.4
3	219.5	3750.8	0.0	210.1	3.2	207.1	9.4	5.3
4	1471.2	57.0	986.0	299.5	1280.5	199.2	187.3	356.4
5	1472.4	59.4	986.0	290.7	357.1	199.2	186.9	356.9
6	0.0	3801.8	1.5	0.0	0.2	0.0	0.0	359.6
7	1474.2	61.5	989.0	97.3	358.7	247.6	188.8	358.4
8	2.0	0.0	4.0	1.7	0.0	1.9	2.5	42.6
9	1475.1	62.2	990.7	99.3	360.6	202.5	190.5	363.9
10	1479.5	5794.0	992.4	102.2	421.0	205.9	194.0	396.8
11	5.3	3.7	8.8	9.0	10.2	5.4	8.8	0.0

Table 4. Delta AIC rankings of the importance of vector, host and transmission parameters explored in mechanistic model combinations (Scenarios as outlined in Table 3) fitted to RRV notifications. Where,  $\beta_{vi}$  the transmission rate from vectors to host species *i*;  $\rho_{\vartheta}$  the phase and amplitude of seasonal transmission;  $\omega_i$  host recrudescence of host species *i*; *v* primary vector population; and  $v_2$  secondary vector population. Host and vector species *i* are defined as kangaroos/wallabies, *k*; possums, *p*; and humans, *h*. Best fitting models ( $\Delta AIC < 4$ ) are in bold, with the most likely model fit having a  $\Delta AIC = 0$ .

https://doi.org/10.1371/journal.ppat.1011944.t004

of transmission scenarios examined, there was minimal model uncertainty in parameters included in the most likely models, with two to three most likely models per site ( $\Delta AIC < 4$ ), except Coorong, which had a single best-fit model (Table 4). Recrudescence was included in the most likely models at only two sites Brisbane and Coorong (Table 4). We find that the models were able to generally capture the number RRV notifications leading to observed RRV outbreaks, having high positive predictive values, but were more variable in capturing the non-outbreak dynamics, having low negative predictive values (Table 5). Modelled RRV notifications in Mildura, Gippsland, and Coorong were captured least well, compared to the other sites in non-outbreak periods (Table 5).

#### Transmission mechanisms

A valuable contribution of this study is the estimation of host, vector, and seasonal parameters associated with the transmission dynamics of RRV in multiple epidemic areas, allowing for an improved understanding of how mechanisms of transmission systems differ among sites. Despite ecological and environmental differences between the epidemic centres examined, we observed minimal variation in the transmission rate estimates between vector and host populations, particularly between vectors and humans (Table 6).

Table 5. Capacity of the mechanistic models to qualitatively capture the dynamics of human RRV incidence. PVO is the positive validated outbreak; and NVO is the negative validated outbreak.

Site	Mean annual incidence per/100,000	Incidence pattern	Observed outbreaks	Modelled outbreak	PVO	NVO
Darwin	99.2	Annual	132	89	0.91	0.37
Brisbane	26.6	Biennial	78	67	0.91	0.21
Mandurah	51.4	Biennial^	82	73	0.90	0.12
Mildura	102.1	Multi-ennial*	43	27	0.92	0.07
Gippsland	32.5	Multi-menial*	32	82	0.92	0.05
Renmark	108.5	Multi-ennial*	35	16	0.89	0.25
Murray Bridge	37.6	Multi-ennial*	22	13	0.94	0.39
Coorong	152.1	Multi-ennial*	29	2	0.89	0

\*On average > 4 years between RRV outbreaks, and ^ On average 1-4 years between RRV outbreaks.

https://doi.org/10.1371/journal.ppat.1011944.t005

Table 6. Weighted model average parameter estimates of vector, nost, and transmission parameters explored in mechanistic model combinations litted to RRV
<b>notifications.</b> Where, $\beta_{vi}$ the transmission rate from vectors to host species <i>i</i> ; $\omega_i$ rate of recrudescence in host species <i>i</i> ; $\epsilon_\mu$ the seasonal phase of transmission; $\sigma_{\varphi}$ the ampli-
tude of seasonal transmission; v <sub>1min</sub> minimum mean estimated vector population in proportion to the maximum observed vector population per mosquito monitoring
trap per week; <i>v</i> <sub>1<i>max</i></sub> estimated maximum mean vector population per mosquito monitoring trap per week; and <i>v</i> <sub>2</sub> estimated mean secondary vector population per mos-
quito monitoring trap per week, $\beta_r$ the false-negative reporting rate (number of unreported infections per one notification), and $\alpha_r$ the false-positive reporting rate (number
of false infections reported per one notification). Host and vector species <i>i</i> are defined as kangaroos/wallabies, <i>k</i> ; possums, <i>p</i> ; and humans, <i>h</i> .

1. 6 Weighted model evenes a neuron stan estimates of vertex heat and two mission neuron stars evaluation

Parameter	Darwin	Brisbane	Mandurah	Mildura	Gippsland	Renmark	Murray Bridge	Coorong
$\beta_{vh}$	$5.24 \times 10^{-4}$	8.89 x 10 <sup>-4</sup>	$7.87 \ge 10^{-4}$	9.65 x 10 <sup>-4</sup>	$3.05 \ge 10^{-4}$	9.76 x 10 <sup>-4</sup>	$9.88 \ge 10^{-4}$	9.99 x 10 <sup>-4</sup>
$\beta_{\nu k}$	$1.24 \ge 10^{-4}$	$1.75 \ge 10^{-4}$	8.15 x 10 <sup>-5</sup>	8.08 x 10 <sup>-6</sup>	4.78 x 10 <sup>-5</sup>	7.16 x 10 <sup>-5</sup>	6.96 x 10 <sup>-7</sup>	9.33 x 10 <sup>-4</sup>
$\beta_{vp}$	$4.40 \ge 10^{-5}$	$1.30 \ge 10^{-4}$	$1.04 \ge 10^{-5}$	6.98 x 10 <sup>-6</sup>	$4.28 \ge 10^{-4}$	$4.41 \ge 10^{-5}$	9.75 x 10 <sup>-6</sup>	$8.17 \ge 10^{-4}$
$\omega_k$	$3.68 \ge 10^{-2}$	$3.60 \ge 10^{-2}$	$2.42 \ge 10^{-3}$	$3.58 \ge 10^{-3}$	9.55 x 10 <sup>-8</sup>	$2.05 \text{ x } 10^{-2}$	$1.15 \ge 10^{-3}$	9.28 x 10 <sup>-2</sup>
$\omega_p$	$3.52 \times 10^{-2}$	2.78 x 10 <sup>-2</sup>	$7.62 \ge 10^{-4}$	5.67 x 10 <sup>-3</sup>	$2.81 \times 10^{-3}$	$2.06 \ge 10^{-3}$	6.36 x 10 <sup>-3</sup>	1.27 x 10 <sup>-8</sup>
$\epsilon_{\mu}$	0	0	5.60 x 10 <sup>-3</sup>	0	0	0	8.18 x 10 <sup>-3</sup>	9.34 x 10 <sup>-1</sup>
$\sigma_{\varphi}$	$4.44 \ge 10^{-01}$	1.00	9.22 x 10 <sup>-02</sup>	$3.00 \ge 10^{-01}$	$4.77 \ge 10^{-01}$	$3.13 \ge 10^{-01}$	2.29 x 10 <sup>-01</sup>	9.34 x 10 <sup>-01</sup>
$v_{1min}$	9.95 x 10 <sup>-3</sup>	$1.00 \ge 10^{-2}$	$1.00 \ge 10^{-2}$	9.98 x 10 <sup>-3</sup>	9.94 x 10 <sup>-3</sup>	9.99 x 10 <sup>-3</sup>	9.99 x 10 <sup>-3</sup>	1.91 x 10 <sup>-3</sup>
v <sub>1max</sub>	16	70	22	34	7	56	21	2
<i>v</i> <sub>2</sub>	5	0	4	2	1089	2	22	35901
$\beta_r$	0.87	0.98	0.93	0.74	1.49 x 10 <sup>-6</sup>	0.74	0.91	$1.02 \ge 10^{-6}$
$\overline{\alpha_r}$	1.19 x 10 <sup>-11</sup>	3.36 x 10 <sup>-12</sup>	8.29 x 10 <sup>-12</sup>	1.07 x 10 <sup>-9</sup>	2.41 x 10 <sup>-9</sup>	1.25 x 10 <sup>-10</sup>	5.04 x 10 <sup>-12</sup>	1.83 x 10 <sup>-9</sup>

https://doi.org/10.1371/journal.ppat.1011944.t006

While seasonality in transmission (representative of vector feeding rate upon hosts) was important in modelling RRV transmission, the phase and amplitude of seasonality and its effect on transmission varied among regions (Table 6). We find seasonal vector feeding commonly amplify RRV transmission across epidemic centres. However, only sites in southern latitudes were found to have regular specific seasons within a year where vector feeding commonly occurs. In Darwin and Brisbane, while there was amplification of RRV transmission resulting from seasonal vector feeding, the timing of this season varied among years, with no common season where vector feeding was more prominent (Table 6).

We found that across sites, for RRV transmission to persist between host and vector populations, the estimated minimum primary vector population generally cannot fall below 0.1% of the maximum observed vector abundance (Table 6). In contrast, the estimated maximum vector population for optimal RRV transmission varied greatly among sites. The primary vector mosquito species chosen for each site were based on the relative species abundance, where greater vector abundance is typically assumed to be a major contributor to viral transmission. However, our analysis showed that despite having a lower relative abundance, in most regions, the secondary vector species still has an important contribution to RRV transmission (Table 6). Despite having a lower relative abundance, in most regions, the secondary vector species still has an important contribution to RRV transmission (Table 6). Despite having a lower relative abundance, in most regions, the secondary vector species still has an important contribution to RRV transmission (Table 6). Despite having a lower relative abundance, in most regions, the secondary vector species still has an important contribution to RRV transmission. This highlights that while the abundance of a vector species is important, the presence of additional secondary vector species, even if less abundant, can still play a key role in transmission dynamics. The consistency of the secondary vector's importance across epidemic regions, which vary in vector species composition, underscores how vector community diversity enables RRV persistence.

Under-reporting (based on false-negative reporting rate) of RRV was estimated to be relatively common across Australia, for every 10 notifications of RRV, there could be between 7.4 to 9.3 additional unreported infections of RRV (i.e., 52–57% of all infections lead to a reported notification), except for in Coorong and Gippsland where the under-reporting rate was less common (Table 6). False-positive notification of RRV was estimated to be very rare in our models and unlikely to influence disease surveillance activities (Table 6). We also noted that the estimated relative abundance of the secondary vector species and under-reporting rate in Gippsland and Coorong was found to be distinctly different from those seen in other sites, with estimates being biologically unrealistic (Table 6). Moreover, models for Gippsland and Coorong had the greatest uncertainty in capturing non-outbreak dynamics. These factors suggest that the parameterisations may not accurately reflect observed RRV transmission at these sites (Table 5).

## Discussion

Despite being critical to the dynamics and control of mosquito-borne diseases, the most important mechanisms driving disease dynamics in humans are often poorly understood. Here we examined the intrinsic mechanisms in the transmission of Ross River virus (RRV), producing human disease dynamics. We investigated the drivers of transmission between varying epidemic regions in Australia using model selection on our mechanistic ODE models. Our models indicate that the transmission of RRV between vectors and hosts is relatively similar across epidemic regions. The importance of multiple vector and host species was supported across all sites. Seasonal influences on vector biting rates for RRV transmission also appears generally important. Our findings further support RRV incidence as being underreported across Australia, with up to 43% of human infections being estimated as undetected/unreported [67]. Economic impacts of RRV on healthcare and lost productivity are estimated to cost approximately \$15 million per annum<sup>1</sup>. However, because of the number of potentially unnotified cases, the true health and economic impact are likely to be more than previously estimated. While underreporting of RRV is expected, we estimate that the false-positive diagnosis of RRV to be infrequent, highlighting the specificity of the current national case definition and laboratory diagnostic procedures.

Zoonotic vector-borne diseases are inherently complex owing to the variety and number of vector and host species that may be involved in pathogen transmission. RRV is no exception, with over 42 known mosquito and 60 vertebrate host species with the potential to contribute to transmission [2,4,6,74]. Intrinsic and extrinsic factors such as vector feeding preferences [4,75,76], relative host competence (i.e., viral titre and viremia), host species community composition [8], host availability to vector mosquitoes, and host abundance, and meteorological and climate conditions [12,13], all contribute to shaping transmission patterns leading to pathogen spillover into humans and mosquito-human-mosquito transmission. This study demonstrates this complexity in transmission dynamics. When deterministically modelling RRV notification, the more complex models that include multiple vectors and host populations and seasonal parameters explain a greater amount of variation in the temporal pattern in RRV transmission than more simplistic transmission scenarios. For example, two vector species, macropod and possum hosts, and seasonality in vector feeding were all supported as important mechanisms for RRV transmission across all eight epidemic sites investigated. In the epidemic centres investigated here, two vector and two reservoir host populations, viral recrudescence in host populations, and seasonality in RRV transmission were assessed in 11 scenario model combinations likely driving transmission. Additional model combinations not considered here may also drive transmission (e.g., other vector or host species). However, the consistency in the best fit models among our sites gives us confidence that these model combinations are a reasonable representation of the transmission dynamics based on the parameters used. This is further supported by minimal variation among our transmission parameter estimates, particularly the vector to human transmission rate ( $\beta_{vh}$ ).

Past investigations of host contributions as RRV reservoirs have generally relied on serological studies detailing relative host competence, which greatly differs among species [1,7,77]. We explored vector-host transmission rates, finding minimal variation in RRV transmission rates between vector and humans ( $\beta_{vh}$ ). Similar patterns in macropod and possum hosts were also observed, with macropod and possum species ( $\beta_{vk}$ ,  $\beta_{vp}$ ) having similar transmission rates to one another among our sites (although more variation than  $\beta_{vh}$ ). This reinforces that, despite the relative competence and abundance of macropods, other species such as possums can contribute to the transmission and amplification of RRV. This supports previous studies, finding that secondary hosts such as possums and host species community compositions may amplify RRV transmission [2,8]. Our analysis showed the secondary vector species contributed to transmission despite having lower relative abundance compared to the primary species. This supports the importance of vector diversity in enabling RRV persistence, rather than abundance of a single dominant species. Moreover, our findings further support urban RRV transmission in built-up areas with few macropods but many possums. Species outside marsupials, such as placental mammals and birds, are increasingly being considered as having the potential to contribute to RRV transmission dynamics<sup>2,4</sup>, and further research to explore their roles would be of value.

We tested the recrudescence of RRV as a likely explanation of the mechanisms involved in RRV persistence, particularly during winter periods when vector populations decline and in southern temperate regions where RRV notifications in humans are infrequent, with multiannual periods between epidemics. In most cases, we find that the viral recrudescence of RRV in hosts was not supported to contribute to RRV transmission. Environmental and seasonal stress is thought to drive immunosuppression allowing for the re-emergence and short-term viremic periods in recovered hosts that may facilitate the reintroduction and circulation of RRV. According to our models, viral recrudescence in hosts likely plays a lesser role in the general transmission of RRV, except for potentially in Coorong and Brisbane.

Generalist feeding mosquitoes display high plasticity in feeding patterns on host species which vary by ecological setting and host community composition and abundance [76,78–80]. Seasonal and environmental factors are associated with vector abundance and are often used in deterministic and predictive modelling for RRV [12,13,81-83]. Our finding on the role of seasonality in vectors transmission, as an indicator of seasonal variation in vector feeding, suggest seasonal amplification commonly occurs across epidemic regions, but that the timing of seasonality in vector feeding is specific to individual regions. Furthermore, for RRV transmission to persist, we found that vector abundance cannot fall below 0.1% of the maximum observed vector abundance. Understanding how small vector populations effectively restrict RRV transmission may prove useful for mosquito control strategies. Because of logistical and cost constraints, mosquito surveillance is often targeted to specific times of the year when disease risk is greatest and for the domestic nuisance caused by mosquito populations. Because of this, vector abundance and its importance in sustaining RRV transmission is relatively unknown for parts of the year when their abundance is lowest. In contrast to common understanding, we also find that the transmission of RRV within our models does not require large population densities of vectors for effective transmission to humans.

Host competence is well documented in serological studies of RRV [55,84]. However, more empirical information on host community structures and host life-history traits associated with transmission are needed to better understand the role that hosts play in RRV transmission. We found transmission rates between vector and host populations were relatively similar across epidemic regions. Where variation between areas was observed, it was largely transmission rates associated with the reservoir hosts and vectors than between vectors and human populations. Here we make assumptions on host densities due to the significant gap in general host abundance data and the lack of temporal surveillance of wildlife host populations. For instance, we assume hosts have relatively stable populations through time, although marsupials

in these regions do undergo population changes [33,39,40,60–62]. Monitoring of host populations could be considered for inclusion in RRV surveillance and monitoring. Notwithstanding these limitations, we bring together the most available realistic combination of host abundance and reproduction likely contributing to transmission. The consistency in the vector to host transmission rates gives confidence they are at least representative of the transmission ecology of RRV.

Our findings suggest seasonal vector feeding influences host-vector transmission, leading to RRV spillover into humans. However, the seasonal phase in which vector feeding contributes to transmission varies across Australia. For instance, the transmission of RRV in Darwin and Brisbane both experience an amplification of RRV from seasonal vector feeding; however, there is no consistent time of year this occurs. We hypothesise, in this instance, areas such as Darwin and to a lesser extent Brisbane, have favourable seasonal and climatic conditions allowing for continuous host breeding, with no distinct breeding season, and with vector populations being present throughout the year with well-defined annual cycles in vector populations. Therefore, RRV transmission can more readily occur year-round. This is in contrast with temperate areas where host and vector populations are heavily driven by environmental patterns spanning several years.

A significant strength of our study is the bringing together of diverse ecological and epidemiological data to describe in depth the mechanistic processes driving RRV transmission. To the best of our knowledge, this is the first study to bring together long-term empirical disease surveillance data to test potential mechanisms driving disease patterns in RRV human incidence. Mechanistic models rely upon understanding several aspects of pathogen, vector, and host interactions to begin determining rates of transmission and the dynamics that underpin these rates. The collection of this heterogeneous data from the literature has limitations [85,86]. For instance, here vertical transmission rates in vector populations are based on laboratory studies conducted on *Aedes camptorhynchus* and are generalised across the vector species used here.

A limitation of our deterministic modelling approach is that measures of uncertainty, such as confidence intervals, are not generated for the parameter estimates. A key outcome this study and our findings is the ability for future research to begin using the parameters estimated here to conduct sensitivity analysis on key parameter important in RRV transmission, such as host and vector populations, to determine the impact varying these populations have on RRV notifications. By conducting this future research, we can begin to assess how many RRV cases are averted by current vector management strategies and evaluate the cost-benefit of varying levels of mitigation. Furthermore, our models do not account for variation in the relative viral titre in hosts. Variation within this parameter may alter transmission. While our models grouped macropods into a single reservoir host category, we incorporated species-level differences in density, viremic period, and birth seasonality based on available empirical evidence. Moreover, we also modelled seasonal macropod births using site-specific data on breeding patterns. These parameterised differences between macropod species and sites capture some of the variability that could drive their transmission roles. Quantifying these variable characteristics between macropod species, and their subsequent impacts on RRV dynamics, represents an important direction for future research. Our current models balance parsimony with incorporating measurable differences between species and sites in density, viremia, and births. But as evidence on other species-specific factors accumulates, models could aim to represent additional sources of variation that may alter transmission. While we attempted to collect the most accurate and precise host population and reproductive estimates specific to each site, these population data represent a static "snapshot". Due to the lack of continuous host population data, our mechanistic models do not adequately capture the fluctuations seen the marsupial

and macropod populations. Here we utilise the most of what host-reservoir population data can be pulled together across our sites. Long term host population surveillance would provide better insight into enzootic transmission, the benefits, logistics and costs associated with regular monitoring would need to be carefully considered. Future research could address this gap by conducting site-specific hosts monitoring programs to inform and improve deterministic and predictive modelling capabilities. Moreover, we further acknowledge that there are other host species that play important roles in RRV transmission. While testing these other hosts was beyond the scope of this study, future research could test other species to better understand their relative role and contribution to RRV circulation. Mechanistic differential equation models allow a large degree of flexibility in parameter estimation and in the fitting of disease distributions despite some limitations. Models used here show that while relying on timepoint estimates of host densities, patterns of disease transmission can still be determined. Mosquito-borne diseases also circulate through multiple vectors and hosts. Here we have simplified transmission with two vector and two host species to prevent overfitting models to obtain transmission rate estimates between these populations. This does not mean to say that additional hosts and vector species not included here do not contribute to RRV transmission. With these new estimated vector and host transmission parameters, future studies can begin to expand and assess further aspects to RRV transmission, including additional vector and host species, and how management practices influence disease dynamics in these epidemic centres.

The modelling presented here accounts for the underreporting of RRV infections, which is an important distinction when understanding true ecological processes in disease transmissions. From an ecological perspective, understanding total disease transmission provides insights into the mechanisms and extent of virus spread between vectors and hosts. However, for practical applications like resource allocation, forecasting reported notifications has greater utility, although uncertainty around underreporting makes predicting true case numbers difficult. Our aim was to investigate RRV transmission dynamics, not develop a forecasting tool per se. We provide a retrospective analysis estimating underreporting rates, which has been lacking for RRV and is valuable ecologically. However, a model directly forecasting notifications would better inform public health planning, albeit with uncertainty about true infections. It is possible that future work might look to use a similar mechanistic modelling approach for forecasting (in addition to the statistical approaches already used), but that would likely need to be after more mechanisms have been explored and a mechanistic basis more settled upon. Approaches separating the infection process from reporting could be beneficial in future work to separately estimate underreporting for ecological understanding and predict notifications for mitigation planning. Investigating how variation in vector control and diagnostics has influenced historical reporting would further improve interpretation of notifications.

In this study we brought together the most diverse set of ecological and epidemiological data for RRV using long-term surveillance to describe the most in-depth understanding of mechanisms driving patterns of human incidence across epidemiologically important areas of Australia. We enhance current understandings of the ecology underlying RRV transmission mechanisms which may be used in public health management. Our results highlight the complex nature in RRV transmission dynamics, and the extent of under-reporting/undiagnosed RRV infections across Australia. The information gained here can be used to inform future research to improve understandings of vector control programs beneficial in reducing health and economic impacts. The establishment of empirically founded mechanistic models fit to empirical data is a critical frontier to effectively evaluate factors influencing mosquito-borne disease dynamics in existing and new epidemic areas.

# **Supporting information**

**S1 Appendix. Vector mosquito population trend.** Mosquito population trends by mosquito species and study site. (DOCX)

**S2** Appendix. R and solver code. Code and solver used to run the ordinary differential equations.

(DOCX)

**S3 Appendix. Transmission scenarios.** Transmission scenarios model combinations for explaining RRV transmission across epidemic centres in Australia. (DOCX)

# Acknowledgments

Mosquito data used this research was supported by Health Departments and Local Councils and the staff within these bodies who have undertaken the mosquito monitoring programs spanning several years. We would like to thank the Departments and staff involved who have contributed to the mosquito surveillance used here, which includes the Arbovirus Surveillance and Research Laboratory at in the School of Pathology and Laboratory Medicine at the University of Western Australia, and the Department of Health, Western Australia; Victorian Department of Health and Human Services; AgriBio Agriculture Victoria; Centre for Disease Control, Northern Territory Department of Health; Brisbane City Council; Department of Environment, Land, Water and Planning, Victoria; Clinical & Health Sciences, University of South Australia. Moreover, RRV host ecology information (such as reproductive timing and population abundance) for Victoria and South Australia was provided by Graeme Maxwell (University of Melbourne) and Allison Mckay.

# **Author Contributions**

Conceptualization: Nicholas Beeton, Scott Carver.

**Data curation:** Iain S. Koolhof, Nicholas Beeton, Katherine Gibney, Peter Neville, Andrew Jardine, Peter Markey, Nina Kurucz, Allan Warchot, Vicki Krause, Michael Onn, Stacey Rowe, Lucinda Franklin, Stephen Fricker, Craig Williams, Scott Carver.

Formal analysis: Iain S. Koolhof, Nicholas Beeton.

- Investigation: Iain S. Koolhof, Silvana Bettiol, Michael Charleston, Simon M. Firestone, Peter Neville, Andrew Jardine, Peter Markey, Nina Kurucz, Allan Warchot, Vicki Krause, Michael Onn, Stacey Rowe, Lucinda Franklin, Stephen Fricker, Craig Williams, Scott Carver.
- Methodology: Iain S. Koolhof, Nicholas Beeton, Michael Charleston, Simon M. Firestone, Scott Carver.

Project administration: Iain S. Koolhof, Silvana Bettiol, Scott Carver.

Software: Iain S. Koolhof.

Supervision: Silvana Bettiol, Michael Charleston, Simon M. Firestone, Scott Carver.

Visualization: Iain S. Koolhof.

Writing - original draft: Iain S. Koolhof, Andrew Jardine, Scott Carver.

Writing – review & editing: Iain S. Koolhof, Nicholas Beeton, Silvana Bettiol, Michael Charleston, Simon M. Firestone, Katherine Gibney, Peter Neville, Andrew Jardine, Peter Markey, Nina Kurucz, Allan Warchot, Vicki Krause, Michael Onn, Stacey Rowe, Lucinda Franklin, Stephen Fricker, Craig Williams, Scott Carver.

#### References

- Harley D, Sleigh A, Ritchie S. Ross River virus transmission, infection, and disease: a cross-disciplinary review. Clin Microbiol Rev 2001; 14:909–932. https://doi.org/10.1128/CMR.14.4.909-932.2001 PMID: 11585790
- 2. Ong OTW, Skinner EB, Johnson BJ, Old JM. Mosquito-Borne Viruses and Non-Human Vertebrates in Australia: A Review. Viruses 2021; 13:265. https://doi.org/10.3390/v13020265 PMID: 33572234
- Walsh MG. Ecological and life history traits are associated with Ross River virus infection among sylvatic mammals in Australia. BMC Ecol 2019; 19:2. <u>https://doi.org/10.1186/s12898-019-0220-5</u> PMID: 30646881
- Stephenson EB, Peel AJ, Reid SA, Jansen CC, McCallum H. The non-human reservoirs of Ross River virus: a systematic review of the evidence. Parasit Vectors 2018; 11:188. <u>https://doi.org/10.1186/</u> s13071-018-2733-8 PMID: 29554936
- Aaskov J, Fokine A, Liu W. Ross River virus evolution: implications for vaccine development. Future Virol 2012; 7:173–178.
- Russell RC. Ross River virus: Ecology and distribution. Annu Rev Entomol 2002; 47:1–31. <u>https://doi.org/10.1146/annurev.ento.47.091201.145100 PMID: 11729067</u>
- Carver S, Bestall A, Jardine A, Ostfeld RS. Influence of Hosts on the Ecology of Arboviral Transmission: Potential Mechanisms Influencing Dengue, Murray Valley Encephalitis, and Ross River Virus in Australia. Vector Borne Zoonotic Dis 2009; 9:51–64. https://doi.org/10.1089/vbz.2008.0040 PMID: 18800866
- Koolhof IS, Carver S. Epidemic host community contribution to mosquito-borne disease transmission: Ross River virus. Epidemiol Infect 2017; 145:656–666. https://doi.org/10.1017/S0950268816002739 PMID: 27890043
- 9. Carver S, Slaney DP, Leisnham PT, Weinstein P. Healthy wetlands, healthy people: mosquito borne disease. *Wetlands and Human Health* Springer, 2015;95–121.
- Potter A, Johansen CA, Fenwick S, Reid SA, Lindsay MDA. The Seroprevalence and Factors Associated with Ross River Virus Infection in Western Grey Kangaroos (Macropus fuliginosus) in Western Australia. *Vector Borne Zoonotic Dis* 2014; 14:740–745. <u>https://doi.org/10.1089/vbz.2014.1617</u> PMID: 25325318
- 11. Potter A, Lindsay MD, MacKenzie JS. Ross River virus-at the interface between humans, animals and the environment. Microbiol Aust 2012; 33:160–162.
- Tall JA, Gatton ML, Tong SL. Ross River Virus Disease Activity Associated With Naturally Occurring Nontidal Flood Events in Australia: A Systematic Review. J Med Entomol 2014; 51:1097–1108. https:// doi.org/10.1603/ME14007 PMID: 26309294
- Koolhof IS, Gibney KB, Bettiol S, Charleston M, Wiethoelter A, Arnold A-L, et al. The forecasting of dynamical Ross River virus outbreaks: Victoria, Australia. Epidemics 2019:100377. https://doi.org/10. 1016/j.epidem.2019.100377 PMID: 31735585
- 14. Cruz-Pacheco G, Esteva L, Vargas C. Seasonality and outbreaks in West Nile virus infection. Bull Math Biol 2009; 71:1378–1393. https://doi.org/10.1007/s11538-009-9406-x PMID: 19288065
- Koolhof IS, Firestone SM, Bettiol S, Charleston M, Gibney KB, Neville PJ, et al. Optimising predictive modelling of Ross River virus using meteorological variables. PLoS Negl Trop Dis 2021; 15:e0009252. https://doi.org/10.1371/journal.pntd.0009252 PMID: 33690616
- Tall JA, Gatton ML. Flooding and arboviral disease: predicting Ross River virus disease outbreaks across inland regions of south-eastern Australia. J Med Entomol 2019; 57:241–251.
- Cutcher Z, Williamson E, Lynch SE, Rowe S, Clothier HJ, Firestone SM. Predictive modelling of Ross River virus notifications in southeastern Australia. Epidemiol Infect 2017; 145:440–450. <u>https://doi.org/ 10.1017/S0950268816002594</u> PMID: 27866492
- Woodruff RE, Guest CS, Gainer MG, Becker N, Lindsay M. Early warning of ross River Virus epidemics —Combining surveillance data on climate and mosquitoes. Epidemiology 2006; 17:569–575. <u>https://doi.org/10.1097/01.ede.0000229467.92742.7b</u> PMID: 16837824
- Weather Reiter P., vector biology, and arboviral recrudescence. The arboviruses: epidemiology and ecology CRC press, 1988;245–256.

- Kermack WO, McKendrick AG. A contribution to the mathematical theory of epidemics. Proc R soc Lond Ser A-Contain Pap Math Phys Character 1927; 115:700–721.
- Dobson A. Population dynamics of pathogens with multiple host species. Am Nat 2004; 164:64–78. https://doi.org/10.1086/424681 PMID: 15540143
- 22. Keeling M, Rohani P. Modeling infectious diseases in humans and animals. Clin Infect Dis 2008; 47:864–866.
- Lindsay M, Oliveira N, Jasinska E, Johansen C, Wright A, Smith D. Western Australia's largest recorded outbreak of Ross River virus disease. Arbovirus Research Aust 1997; 7:147–152.
- Lindsay M, Broom A, Oliveira N, Jasinska E, van Heuzen B, Caulfield S, et al. Western Australian arbovirus surveillance and research program, Annual Report: 1997–1998. Western Australia, Australia: Arbovirus Surveillance and Research Laboratory and the Western Australian Centre for Pathology and Medical Research 1998.
- Denholm L, Beeton NJ, Forbes LK, Carver S. A model for the dynamics of Ross River Virus in the Australian environment. Lett Biomath 2017; 4:187–206.
- 26. Koolhof I, Bettiol S, Carver S. Fine-temporal forecasting of outbreak probability and severity: Ross River virus in Western Australia. Epidemiol Infect 2017; 145:2949–2960. <u>https://doi.org/10.1017/S095026881700190X</u> PMID: <u>28868994</u>
- Bi P, Hiller JE, Cameron AS, Zhang Y, Givney R. Climate variability and Ross River virus infections in Riverland, South Australia, 1992–2004. Epidemiol Infect 2009; 137:1486–1493. <u>https://doi.org/10.</u> 1017/S0950268809002441 PMID: 19296873
- Bedoya-Pérez M, Lawes M, McMahon C. Protecting vulnerable land from high wallaby densities Stage
   Meat & Livestock Australia Limited, 2017.
- Croft D. Socio-Ecology of the Antilopine Wallaroo, Macropus-Antilopinus, in the Northern-Territory, With Observations on Sympatric Macropus-Robustus-Woodwardii and Macropus-Agilis. Wildl Res 1987; 14:243–255.
- Kerle J. The population dynamics of a tropical possum, Trichosurus vulpecula arnhemensis Collett. Wildl Res 1998; 25:171–181.
- **31.** Stirrat SC. Age structure, mortality and breeding in a population of agile wallabies (Macropus agilis). Aust J Zool 2008; 56:431–439.
- **32.** Australian Bureau of Statistics. Data by Region. 2016 [cited on 2016 Feb 2]. Available from: <a href="https://itt.abs.gov.au/itt/r.jsp?databyregion">https://itt.abs.gov.au/itt/r.jsp?databyregion</a>.
- Caughley G, Grigg GC. Numbers and distribution of kangaroos in the Queensland pastoral zone. Wildl Res 1982; 9:365–371.
- Rosen L, Gubler DJ, Bennett PH. Epidemic polyarthritis (Ross River) virus infection in the Cook Islands. Am J Trop Med Hyg 1981; 30:1294–1302. https://doi.org/10.4269/ajtmh.1981.30.1294 PMID: 7325286
- Dhileepan K, Azuolas J, Gibson C. Evidence of vertical transmission of Ross River and Sindbis viruses (Togaviridae: Alphavirus) by mosquitoes (Diptera: Culicidae) in southeastern Australia. J Med Entomol 1996; 33:180–182. https://doi.org/10.1093/jmedent/33.1.180 PMID: 8906927
- 36. Moloney PD, Ramsey DS, Scroggie MP. A state-wide aerial survey of kangaroos in Victoria. Environment, Land, Water and Planning: Victoria State Government, 2017.
- Viggers K, Lindenmayer D. A population study of the mountain brushtail possum (Trichosurus caninus) in the central highlands of Victoria. Aust J Zool 2000; 48:201–216.
- Wilson ME, Coulson G, Shaw G, Renfree MB. Deslorelin implants in free-ranging female eastern grey kangaroos (Macropus giganteus): mechanism of action and contraceptive efficacy. Wildl Res 2013; 40:403–412.
- **39.** Cairns S, Grigg G. Population dynamics of red kangaroos (Macropus rufus) in relation to rainfall in the South Australian pastoral zone. *Journal of Applied Ecology* 1993:444–458.
- 40. Caughley G, Grigg G. Surveys of the distribution and density of kangaroos in the pastoral zone of South Australia, and their bearing on the feasibility of aerial survey in large and remote areas. Wildl Res 1981; 8:1–11.
- 41. Niebylski M, Craig JG. Dispersal and survival of Aedes albopictus at a scrap tire yard in Missouri. J Am Mosq Control Assoc 1994; 10:339–343. PMID: 7807074
- 42. Australian Bureau of Statistics. 3218.0—Regional Population Growth, Australia, 2013–14. 2016 [cited on 2016 Feb 2]. Available from: http://www.abs.gov.au/AUSSTATS/abs@.nsf/DetailsPage/3218. 02013-14?OpenDocument.
- 43. Carver S, Spafford H, Storey A, Weinstein P. Dryland Salinity and the Ecology of Ross River Virus: The Ecological Underpinnings of the Potential for Transmission. Vector-Borne Zoonotic Dis 2009; 9:611–622. https://doi.org/10.1089/vbz.2008.0124 PMID: 19326966

- Canyon D, Hii J, Muller R. Effect of diet on biting, oviposition, and survival of Aedes aegypti (Diptera: Culicidae). J Med Entomol 1999; 36:301–308. https://doi.org/10.1093/jmedent/36.3.301 PMID: 10337099
- Carron A, Bichaud L, Platz N, Bicout D. Survivorship characteristics of the mosquito Aedes caspius adults from southern France under laboratory conditions. Med Vet Entomol 2008; 22:70–73. https://doi. org/10.1111/j.1365-2915.2008.00718.x PMID: 18380656
- Harrington LC, Buonaccorsi JP, Edman JD, Costero A, Kittayapong P, Clark GG, et al. Analysis of survival of young and old Aedes aegypti (Diptera: Culicidae) from Puerto Rico and Thailand. Med Vet Entomol 2001; 38:537–547.
- 47. Leisnham PT, Sala L, Juliano SA. Geographic variation in adult survival and reproductive tactics of the mosquito Aedes albopictus. Med Vet Entomol 2014; 45:210–221.
- Muir LE, Kay BH. Aedes aegypti survival and dispersal estimated by mark-release-recapture in northern Australia. Am J Trop Med Hyg 1998; 58:277–282. <u>https://doi.org/10.4269/ajtmh.1998.58.277</u> PMID: 9546403
- 49. Watson TM, Saul A, Kay BH. Aedes notoscriptus (Diptera: Culicidae) survival and dispersal estimated by mark-release-recapture in Brisbane, Queensland, Australia. Med Vet Entomol 2000; 37:380–384. https://doi.org/10.1093/jmedent/37.3.380 PMID: 15535581
- Glass K. Ecological mechanisms that promote arbovirus survival: a mathematical model of Ross River virus transmission. *Trans R Soc Trop Med Hyg* 2005; 99:252–260. <u>https://doi.org/10.1016/j.trstmh.</u> 2004.08.004 PMID: 15708384
- Lindsay MDA, Broom AK, Wright AET, Johansen CA, Mackenzie JS. Ross River virus isolations from mosquitos in arid regions of Western-Australia—implication of vertical transmission as a means of persistence of the virus. *Am J Trop Med Hyg* 1993; 49:686–696.
- Chitnis N, Hyman JM, Manore CA. Modelling vertical transmission in vector-borne diseases with applications to Rift Valley fever. J Biol Dyn 2013; 7:11–40. https://doi.org/10.1080/17513758.2012.733427 PMID: 23098257
- 53. Kay B, Aaskov J. Ross River virus (epidemic polyarthritis). The arboviruses: epidemiology and ecology 1989; 4:93–112.
- Choi YH, Comiskey C, Lindsay MDA, Cross JA, Anderson M. Modelling the transmission dynamics of Ross River virus in Southwestern Australia. Ima Journal of *Math Med Biol* 2002; 19:61–74. PMID: 12408224
- 55. Boyd AM, Hall RA, Gemmell RT, Kay BH. Experimental infection of Australian brushtail possums, Trichosurus vulpecula (Phalangeridae: Marsupialia), with Ross River and Barmah Forest viruses by use of a natural mosquito vector system. Am J Trop Med Hyg 2001; 65:777–782. <u>https://doi.org/10.4269/ ajtmh.2001.65.777 PMID: 11791974</u>
- 56. Boyd AM. Interactions between common vertebrate hosts and the mosquito vectors of Ross River and Barmah Forest viruses in urban Brisbane, south east Queensland, Australia. 2004.
- 57. Kerle JA, Howe CJ. The Breeding Biology of a Tropical Possum, Trichosurus-Vulpecula-Arnhemensis (Phalangeridae, Marsupialia). Aust J Zool 1992; 40:653–665.
- Eymann J, Cooper DW, Herbert CA. Brushtail possums (Trichosurus vulpecula) in metropolitan Sydney: population biology and response to Suprelorin contraceptive implants. Aust J Zool 2013; 61:78–86.
- **59.** Poole W. Breeding in the grey kangaroo, Macropus giganteus, from widespread locations in eastern Australia. Wildl Res 1983; 10:453–466.
- **60.** Quin D. Age structures, reproduction and mortality of the eastern grey kangaroo (Macropus giganteus Shaw) from Yan Yean, Victoria. Kangaroos, Wallabies and Rat-kangaroos'.(Eds Grigg G., Jarman P. and Hume I.) *pp* 1989:787–794.
- Bayliss P. The population dynamics of red and western grey kangaroos in arid New South Wales, Australia. I. Population trends and rainfall. J Anim Ecol 1985; 1:111–125.
- Cairns S, Grigg G, Beard L, Pople A, Alexander P. Western grey kangaroos, Macropus fuliginosus, in the South Australian pastoral zone: populations at the edge of their range. Wildl Res 2000; 27:309–318.
- **63.** Mayberry C, Maloney SK, Mawson P, Bencini R. Seasonal anoestrus in western grey kangaroos (Macropus fuliginosus ocydromus) in south-western Australia. Aust Mammal 2010; 32:189–196.
- How RA, Hillcox SJ. Brushtail possum, Trichosurus vulpecula, populations in south-western Australia: demography, diet and conservation status. Wildl Res 2000; 27:81–89.
- Victorian Department of Environment, Land, Water, and Planning. The Common Brushtail Possum is the most abundant and familiar of the Australian possums. 2018 [cited on 2018 June 18]. Available from: <u>https://www.wildlife.vic.gov.au/\_\_data/assets/pdf\_file/0025/92437/Common-Brushtail-Possum.</u> pdf.

- 66. WestleyWise VJ, Beard JR, Sladden TJ, Dunn TM, Simpson J. Ross river virus infection on the North Coast of New South Wales. Aust N Z J Public Health 1996; 20:87–92. <u>https://doi.org/10.1111/j.1467-842x.1996.tb01343.x PMID: 8799074</u>
- Aaskov J, Fraser J, Dalglish DA. Specific and non-specific immunological changes in epidemic polyarthritis patients. Aust J Exp Biol Med Sci 1981; 59:599–608. <u>https://doi.org/10.1038/icb.1981.52</u> PMID: 7325890
- Gibbons CL, Mangen M-JJ, Plass D, Havelaar AH, Brooke RJ, Kramarz P, et al. Measuring underreporting and under-ascertainment in infectious disease datasets: a comparison of methods. BMC Public Health 2014; 14:147–154. https://doi.org/10.1186/1471-2458-14-147 PMID: 24517715
- Selvey L, Donnelly J, Lindsay M, PottumarthyBoddu S, D'Abrera V, Smith D. Ross River virus infection surveillance in the Greater Perth Metropolitan area—has there been an increase in cases in the winter months? Communicable Diseases Intelligence—Quarterly Report 2014; 38:E114–22. PMID: 25222206
- 70. Burnham KP, Anderson DR. Model Selection and Multimodel Inference: A Practical Information-theoretic Approach. Vol. 2nd ed. New York: Springer, 2002.
- Symonds MR, Moussalli A. A brief guide to model selection, multimodel inference and model averaging in behavioural ecology using Akaike's information criterion. Behav Ecol Sociobiol 2011; 65:13–21.
- 72. Grolemund G, Wickham H. Dates and Times Made Easy with lubridate. J Stat Softw 2011; 40:1–25.
- Soetaert K, Petzoldt T, Setzer RW. Solving Differential Equations in R: Package deSolve. J Stat Softw 2010; 33:1–25.
- 74. Claflin SB, Webb CE. Ross River Virus: Many Vectors and Unusual Hosts Make for an Unpredictable Pathogen. PLoS Pathog 2015; 11:1–5. https://doi.org/10.1371/journal.ppat.1005070 PMID: 26335937
- Lyimo IN, Ferguson HM. Ecological and evolutionary determinants of host species choice in mosquito vectors. Trends Parasitol 2009; 25:189–196. https://doi.org/10.1016/j.pt.2009.01.005 PMID: 19269900
- 76. Takken W, Verhulst NO. Host preferences of blood-feeding mosquitoes. Annu Rev Entomol 2013; 58:433–453. https://doi.org/10.1146/annurev-ento-120811-153618 PMID: 23020619
- 77. Boyd A, Kay BH. Solving the urban puzzle of Ross River and Barmah Forest viruses. Arbrovirus Research in Australia 2001; 8:14–22.
- 78. Flies EJ, Flies AS, Fricker SR, Weinstein P, Williams CR. Regional comparison of mosquito bloodmeals in South Australia: implications for Ross River virus ecology. J Med Entomol 2016; 53:902–910. <u>https:// doi.org/10.1093/jme/tjw035</u> PMID: 27113100
- 79. LoGiudice K, Duerr STK, Newhouse MJ, Schmidt KA, Killilea ME, Ostfeld RS. Impact of host community composition on Lyme disease risk. Ecology 2008; 89):2841–2849. <u>https://doi.org/10.1890/07-1047.1</u> PMID: 18959321
- Zeilinger AR, Daugherty MP. Vector preference and host defense against infection interact to determine disease dynamics. *Oikos* 2014; 123:613–622.
- Jacups SP, Carter J, Kurucz N, McDonnell J, Whelan PI. Determining meteorological drivers of salt marsh mosquito peaks in tropical northern Australia. J Vector Ecol 2015; 40:277–281. <u>https://doi.org/ 10.1111/jvec.12165 PMID: 26611962</u>
- Jacups SP, Whelan PI, Currie BJ. Ross River virus and Barmah Forest virus infections: A review of history, ecology, and predictive models, with implications for tropical northern Australia. Vector Borne Zoonotic Dis 2008; 8:283–297. https://doi.org/10.1089/vbz.2007.0152 PMID: 18279007
- Jacups SP, Whelan PI, Markey PG, Cleland SJ, Williamson GJ, Currie BJ. Predictive indicators for Ross River virus infection in the Darwin area of tropical northern Australia, using long-term mosquito trapping data. Trop Med Int Health 2008; 13:943–52. <u>https://doi.org/10.1111/j.1365-3156.2008.02095.x</u> PMID: 18482196
- Kay B, Hall R, Fanning I, Mottram P, Young P, Pollitt C. Experimental infection of vertebrates with Murray Valley encephalitis and Ross River viruses. Arbovirus Research in Australia. Proceedings—Symposium, 1986.
- Mordecai EA, Caldwell JM, Grossman MK, Lippi CA, Johnson LR, Neira M, et al. Thermal biology of mosquito-borne disease. Ecology Letters 2019; 22:169–1708. <u>https://doi.org/10.1111/ele.13335</u> PMID: 31286630
- Mordecai EA, Paaijmans KP, Johnson LR, Balzer C, Ben-Horin T, de Moor E, et al. Optimal temperature for malaria transmission is dramatically lower than previously predicted. Ecology letters 2013; 16:22–30. https://doi.org/10.1111/ele.12015 PMID: 23050931